Author's response to reviews

Title: Handheld computers and the 21st century surgical team: a pilot study

Authors:

Omer Aziz (omer.aziz@btinternet.com)
Sukhmeet S Panesar (sukhmeet.panesar@imperial.ac.uk)
Gopalakrishnan Netuveli (Gnetuveli@aol.com)
Paraskevas Paraskeva (p.paraskevas@imperial.ac.uk)
Ara Darzi (a.darzi@imperial.ac.uk)
Aziz Sheikh (aziz.sheikh@ed.ac.uk)

Version: 4 Date: 28 July 2005

Author's response to reviews:

Dear Editor,

Thank you very much for your letter requesting final formatting changes to this manuscript which have now been made. I would also like to respond to the three questions you have asked:

1. In the past five years have you received reimbursements, fees, funding, or salary from an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? Is such an organization financing this manuscript (including the article-processing charge)? If so, please specify.

   NO

2. Do you hold any stocks or shares in an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? If so, please specify.

   NO

3. Do you hold or are you currently applying for any patents relating to the content of the manuscript? Have you received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript? If so, please specify.

   NO

4. Do you have any other financial competing interests? If so, please specify.

   NO

5. Are there any non-financial competing interests (political, personal, religious, academic, intellectual, commercial or any other) to declare in relation to this manuscript? If so, please specify.

   NO

Please do not hesitate to contact me if there is any other information that you may require.

Yours sincerely

Omer Aziz
Corresponding author